Author’s response to reviews

Title: Bilateral acute angle-closure glaucoma following Tramadol subcutaneous administration

Authors:

Anis Mahmoud (mahmoudanisss@hotmail.fr)
Fatma Abid (fatmabid@gmail.com)
Imen Ksiaa (khay.imen@yahoo.fr)
Sourour Zina (sourourzina@hotmail.com)
Riadh Messaoud (messoud_riadh@yahoo.fr)
Moncef Khairallah (moncef.khairallah@yahoo.fr)

Version: 1 Date: 27 Oct 2017

Author’s response to reviews:

Dear Editors, Dear Reviewers,

Thank you for allowing us to revise the manuscript BOPH-D-17-00537 entitled “Bilateral acute angle closure glaucoma following tramadol subcutaneous administration” we submitted for publication in BMC Ophthalmology.

We thank the reviewers for their constructive and useful comments. We made amendments to the manuscript accordingly. Specific responses to each of the comments are provided below.

Sincerely yours,

Moncef Khairallah, MD

Reviewers’ comments:

Reviewer #1:

Reviewer’s comments:

Page 5, Line 26: Grammatical error.
Authors’ responses:

Thank you for the suggestion. The correction was made.

Reviewer’s comments:

It would be really interesting if authors could add an image after iridotomy, either gonioscopy or AS-OCT to document the complete angle opening.

Authors’ responses:

Thank you for the suggestion. A control image was added.

Reviewer #2:

Reviewer’s comments:

Authors should check the format and consistency of references.

Authors’ responses:

Thank you for the suggestion. Improvement of references list was made.